Author's response to reviews

Title: Determining the Validity and Reliability of the Chinese Version of the Kidney Disease Quality of Life Questionnaire (KDQOL-36TM)

Authors:

Xingjuan Tao (tao.xingjuan@connect.polyu.hk)
Susan Ka Yee Chow Dr (susan.chow@polyu.edu.hk)
Frances KY Wong Prof (frances.wong@polyu.edu.hk)

Version: 2 Date: 22 March 2014

Author's response to reviews: see over
Dear editor,

Hello. I attached a paper titled “Validity and Reliability Estimates of the Chinese Version of the Kidney Disease Quality of Life Questionnaire (KDQOL-36™)” for your considerations. The article type is Research articles. The brief version of KDQOL-36™ consisted of 36 items and found to be suitable for use in routine evaluations of quality of care in busy practice settings and has been used extensively to evaluate CKD patients in different stages. A valid Chinese version of the KDQOL-36™ will enable researchers to examine the quality of life of CKD patients within the country and allow for cross-country comparisons to be made. The paper was completed according to the guidelines required by BMC Nephrology and has not been submitted to other journals with similar content or data set.

Thank you and I look forward to your favorable reply.

Yours sincerely,

Susan Chow

Susan Chow (Dr)
Assistant Professor
Programme Leader-Master of Nursing
School of Nursing
The Hong Kong Polytechnic University